Author’s response to reviews

Title: High APACHE 2 score and long length of bowel resection impair the outcomes in patients with necrotic bowel induced hepatic portal venous gas

Authors:

Jin-Ming Wu (kptkptkpt@yahoo.com.tw)
Ming-Shian Tsai (mstsai@ntuh.gov.tw)
Ming-Tsan Lin (linmt@ntu.edu.tw)
Yu-Wen Tien (ywtien5106@ntu.edu.tw)
Tzu-Hsin Lin (jsl555@gmail.com)

Version: 3 Date: 5 March 2011

Author’s response to reviews: see over
Dear editor in chief,

We listed the sentences corrected below.

A. First page
1. We changed the affiliation of 2nd author, Ming-Shian Tsai to Division of General Surgery, Department of Surgery, E-Da Hospital, Kaohsiung, Taiwan.
2. We add the city, and country in the affiliation of the other authors.

B. Third page
1. We add the "Backgrounds" in the beginning of the manuscript.
2. We correct the structure of the manuscript according to your suggestions.

Thanks for accepting our paper. Sincerely,

Jin-Ming Wu, M.D. NTUH